Title: E7080, a multi-targeted tyrosine kinase inhibitor suppresses tumour cell migration and invasion

Authors:

Hilary Glen (h.glen@beatson.gla.ac.uk)
Susan Mason (s.mason@beatson.gla.ac.uk)
Hitesh Patel (h.patel@ed.ac.uk)
Kenneth Macleod (kenneth.macleod@ed.ac.uk)
Valerie G Brunton (v.brunton@ed.ac.uk)

Version: 3 Date: 20 July 2011

Author's response to reviews:

Below is a point by point description of the changes made to the final version

1. Author list has been amended and is now the same as in the submission form
2. Country affiliations have been included
3. Author contributions now include a statement that all authors have read and approved the final version
4. Highlighting has been removed from the text
5. Vertical lines have been removed from Table 1
6. The figures have been cropped and the figure number removed from each
7. The manuscript has been checked for typographical errors